**Author's response to reviews**

**Title:** Urological research in sub-Saharan Africa: a retrospective cohort study of abstracts presented at the Nigerian Association of Urological Surgeons Conferences

**Authors:**

Jibril O Bello (jabarng@yahoo.com)

**Version:** 2  **Date:** 9 March 2013

**Author's response to reviews:**

Title: An improvement in the wording of the title of the manuscript was done with the aim to better capture the essence of the research.

Abstract: The abstract was improved on in the result section by inclusion of significant result and the p values.

Main text of Manuscripts: Improvements were made in grammar. Improvements were also made in the style of reporting (arrangement of) the results in the results section of the manuscript.